RESEARCH ARTICLE

Zika virus outbreak in the Pacific: Vector competence of regional vectors

Elodie Calvez¹, Laurence Mousson², Marie Vazeille², Olivia O'Connor¹, Van-Mai Cao-Lormeau³, Françoise Mathieu-Daudé⁴, Nicolas Pocquet⁵, Anna-Bella Failloux^{2‡}, Myrielle Dupont-Rouzeyrol^{1‡}*

 Institut Pasteur de Nouvelle-Calédonie, URE-Dengue et autres Arboviroses, Nouméa, New Caledonia,
 Institut Pasteur, Arboviruses and Insect Vectors Laboratory, Paris, France, 3 Unit of Emerging Infectious Diseases, Institut Louis Malarde, French Polynesia & Aix Marseille Univ, IRD, AP-HM, SSA, VITROME, IHU-Méditérranée Infection, Marseille, France, 4 Institut de Recherche pour le Développement, UMR MIVEGEC IRD, CNRS, UM, Montpellier, France, 5 Institut Pasteur de Nouvelle-Calédonie, URE-Entomologie Médicale, Nouméa, New Caledonia

‡ These authors are joint senior authors on this work.
* mdupont@pasteur.nc

Abstract

Background

In 2013, Zika virus (ZIKV) emerged in French Polynesia and spread through the Pacific region between 2013 and 2017. Several potential *Aedes* mosquitoes may have contributed to the ZIKV transmission including *Aedes aegypti*, the main arbovirus vector in the region, and *Aedes polynesiensis*, vector of lymphatic filariasis and secondary vector of dengue virus. The aim of this study was to analyze the ability of these two Pacific vectors to transmit ZIKV at a regional scale, through the evaluation and comparison of the vector competence of wild *Ae. aegypti* and *Ae. polynesiensis* populations from different Pacific islands for a ZIKV strain which circulated in this region during the 2013–2017 outbreak.

Methodology/principal findings

Field *Ae. aegypti* (three populations) and *Ae. polynesiensis* (two populations) from the Pacific region were collected for this study. Female mosquitoes were orally exposed to ZIKV (10⁷ TCID50/mL) isolated in the region in 2014. At 6, 9, 14 and 21 days post-infection, mosquito bodies (thorax and abdomen), heads and saliva were analyzed to measure infection, dissemination, transmission rates and transmission efficiency, respectively. According to our results, ZIKV infection rates were heterogeneous between the *Ae. aegypti* populations, but the dissemination rates were moderate and more homogenous between these populations. For *Ae. polynesiensis*, infection rates were less heterogeneous between the two populations tested. The transmission rate and efficiency results revealed a low vector competence for ZIKV of the different *Aedes* vector populations under study.

Conclusion/significance

Our results indicated a low ZIKV transmission by *Ae. aegypti* and *Ae. polynesiensis* tested from the Pacific region. These results were unexpected and suggest the importance of other



Citation: Calvez E, Mousson L, Vazeille M, O'Connor O, Cao-Lormeau V-M, Mathieu-Daudé F, et al. (2018) Zika virus outbreak in the Pacific: Vector competence of regional vectors. PLoS Negl Trop Dis 12(7): e0006637. https://doi.org/10.1371/ journal.pntd.0006637

Editor: Benjamin Althouse, Institute for Disease Modeling, UNITED STATES

Received: February 22, 2018

Accepted: June 25, 2018

Published: July 17, 2018

Copyright: © 2018 Calvez et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: All relevant data are within the paper and its Supporting Information files.

Funding: This work was supported by the Actions Concertées Inter Pasteuriennes (ZikAe Project, ACIP A-15-2014). This work was partly funded by the European Union's Horizon 2020 Research and Innovation Programme under ZIKAlliance Grant Agreement No. 734548 and by the Ministère des Outre-mer (SEOM). The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

factors especially the vector density, the mosquito lifespan or the large immunologically naive fraction of the population that may have contributed to the rapid spread of the ZIKV in the Pacific region during the 2013–2017 outbreak.

Author summary

Zika virus (ZIKV) was isolated for the first time in Uganda in 1947. The virus was sporadically detected in Africa and Asia during few decades. The first human outbreak was declared in Yap Island in Micronesia in 2007. The virus reemerged in 2013 in French Polynesia and circulated throughout the Pacific region and in Americas between 2013 and 2017. ZIKV is transmitted to human by the bite of an infected mosquito. *Aedes* mosquitoes were particularly incriminated in the ZIKV outbreak. In this study, we showed for the first time the results of vector competence study of wild *Aedes aegypti* and *Aedes polynesiensis* from different islands of the Pacific region. Our results demonstrated an unexpected homogenous and low ZIKV vector competence for the different populations of these two vectors. These findings seemed to indicate that other factors have contributed to the rapid spread of ZIKV in the Pacific region.

Introduction

Zika fever is an emerging vector borne disease caused by a single stranded RNA virus, Zika virus (ZIKV) belonging to the genus *Flavivirus* [1]. ZIKV is transmitted to humans by the bite of infected vectors mainly *Aedes* mosquitoes [2–4]. Since its first isolation in Uganda in 1947, ZIKV has been detected occasionally in Africa and Asia [5]. The first human outbreak was declared in Yap state in Federated States of Micronesia in 2007 [4]. ZIKV reemerged in the Pacific region in 2013 in French Polynesia and then spread across the region and to the Americas between 2014 and 2016 [5–9]. This recent outbreak was associated with neurological disorders as Guillain-Barré syndrome [10–12], myasthenia gravis [13] and microcephaly [14, 15] were reported in the Pacific region and in South and Central America including the Caribbean (http://www.paho.org). Phylogenetic studies indicated that ZIKV is divided into two lineages: African and Asian with less than 12% of genetic divergence [16, 17]. The 2013–2016 outbreak was due to the emergence of ZIKV belonging to Asian lineage, and genetic analysis revealed divergence between the Pacific and the American clades [18, 19].

During the 2013 French Polynesian ZIKV outbreak, more than 8,700 suspected cases and 30,000 medical consultations were reported by the sentinel surveillance network [10]. A recent seroprevalence study estimated that more than half of the population was infected by ZIKV during the outbreak in this territory [20]. From French Polynesia, the disease spread to New Caledonia in 2013 [7, 21] affecting the whole territory. Based on the number and proportion of confirmed cases and the proportion of ZIKV infections among arboviral syndromes recorded in the population, New Caledonia Health Authorities estimated the number of cases at about 11,000 [11]. From 2014 to 2017, ZIKV was detected in the Cook Islands, Vanuatu, Fiji, Samoa, Salomon Islands, Tonga and American Samoan (http://www.spc.int/phd/epidemics/) [14, 22].

In the Pacific landscape, different mosquito species are present and some of them are known as potential arbovirus vectors, especially mosquitoes from *Aedes* genus. *Ae. aegypti* is present in most Pacific islands with few exceptions [23]. The presence of *Ae. albopictus* is confirmed in 5 out of 17 countries and territories of this region and its expansion through the Western Pacific

islands is observed right up to Fiji and Tonga islands [24, 25]. *Ae. polynesiensis* distribution is reported in the Eastern part of the Pacific, from French Polynesia to Fiji [22]. In addition, the presence of other vectors is recorded in some local parts of the Pacific region as *Aedes hensilli*, *Aedes scutellaris* and other *Aedes* species belonging to the scutellaris group ([26–28].

Previous studies evaluated the vector competence of *Aedes* vectors for ZIKV belonging to both lineages. Mosquitoes from Africa [29], Asia [30, 31] and America [32, 33] were infected by ZIKV strains belonging to the African lineage and showed various results. The vector competence for ZIKV belonging to the Asian lineage was also evaluated with vectors from America [2, 34] Australia [35] and Europe [36]. However, the ability of the vectors from the Pacific to transmit ZIKV has been poorly studied. *Ae. hensilli* from Yap island, appeared to be able to disseminate African ZIKV strain but the transmission has not been evaluated. However this mosquito was the most collected species in Yap island, Federated States of Micronesia, during the ZIKV outbreak in 2007 (41%) supporting its possible role as a vector during the outbreak [27]. Laboratory strains of *Ae. aegypti* and *Ae. polynesiensis* from French Polynesia were shown to be able to get infected and disseminate an Asian/Pacific ZIKV strain isolated in French Polynesia in 2013. Furthermore, infectious ZIKV particles have been found in saliva of *Ae. aegypti* from French Polynesia as soon as 6 days post-infection [37].

The Pacific landscape promotes specific environmental conditions that have influenced the genetic of the main arboviruses vector *Ae. aegypti* [22, 38]. The arbovirus transmission depends on the specific combination of mosquito and virus genotypes [39–41]. At a regional scale, this specific interaction has already been studied for dengue virus and highlighted significant transmission differences between New Caledonian and Polynesian *Ae. aegypti* [42]. To our knowledge, no studies have compared the vector competence of several field *Aedes* species from the Pacific region with an Asian/Pacific ZIKV strain that circulated in this region. As the Pacific was at the origin of the ZIKV pandemic, there is a real interest in investigating the interactions between arbovirus and Pacific vectors at a regional scale.

For this purpose, we collected three field populations of *Ae. aegypti* in New Caledonia (West Pacific), Samoa (Center) and French Polynesia (East) in order to evaluate and compare the vector competence of several populations of the main Pacific vector. Additionally, we collected two field populations of *Ae. polynesiensis* in Wallis (from the Territory of Wallis and Futuna Islands) (Center) and Tahiti (French Polynesia) (East) to evaluate the vector competence of this potential regional vector and to compare the results obtained for these two *Aedes* vectors. We performed these vector competence studies with the ZIKV Pacific strain which circulated in New Caledonia in 2014 and belongs to the Asian/Pacific clade.

Materials and methods

Mosquito collections

Mosquitoes were sampled in five sites in the Pacific region in 2016 with permission of the residents and they were shipped to the Laboratory of Arboviruses and Insect Vectors (AIV) at the Institut Pasteur, Paris (Table 1 and Fig 1). The adults were maintained at 28°C and 80% of humidity with a 16:8h light:dark cycle and fed with a 10% sucrose solution *ad libitum*. Females were blood-fed several times to obtain the F1-F3 generation of mosquitoes used for infection assays.

Viral strain

The viral strain used in this study was a ZIKV isolated from a patient' serum in New Caledonia in April 2014 (NC-2014-5132) (GenBank SRR5309451) (Collection of the Institut Pasteur de Nouvelle-Calédonie). Viral stocks were prepared after five passages on African green monkey

Sample name	Country	Species	Sampled stage	Generation used in laboratory experiments
Aae-French Polynesia	French Polynesia	Ae. aegypti	Larvae / Pupae (<300)	F ₁
Aae-New Caledonia	New Caledonia	Ae. aegypti	Larvae / Pupae (<200)	F ₂
Aae-Samoa	Samoa	Ae. aegypti	Egg	F ₃
Apo-French Polynesia	French Polynesia	Ae. polynesiensis	Larvae / Pupae (<300)	F ₁
Apo-Wallis	Wallis and Futuna	Ae. polynesiensis	Adult (160-180 females)	F ₃

Table 1. *Aedes aegypti* and *Aedes polynesiensis* populations sampled in the Pacific region, 2016. When available, data regarding the number of individuals sampled are provided in brackets.

https://doi.org/10.1371/journal.pntd.0006637.t001

kidney cells (Vero E6, ATCC cell lines) maintained in DMEM medium (Gibco) supplemented with 10% fetal bovine serum (FBS) (Gibco). Supernatant were collected after 7 days of incubation at 37°C. The titration of the viral stock was performed by serial 10-fold dilutions on Vero cells and expressed in $TCID_{50}/mL$.

Mosquito oral infections

For each mosquito population, 4-5 boxes of 60 five to seven day-old females, not previously blood-fed, were starved for 24 hours before infection. They were allowed to take an infectious blood meal, through a capsule (Hemotek system) covered by a pig intestine (obtained from a commercial purchased pig intestine) as membrane and containing 2 ml of washed rabbit erythrocytes obtained directly from a rabbit (New Zeland white Rabbit, Charles River) and 1 ml of viral suspension supplemented with adenosine triphosphate at 5 mM. The ZIKV concentration in the blood meal was 10^7 TCID₅₀/mL. After the blood meal, fully engorged females were transferred into new containers and maintained at 28°C and 80% of humidity under a 12:12h light-dark cycle with free access to a 10% sucrose solution.



Fig 1. Pacific map locating *Ae. aegypti* and *Ae. polynesiensis* sampling sites. *Ae. aegypti* sampling sites are represented by the red stars and *Ae. polynesiensis* sampling sites by the dark star in a white dot. This map was generated using map files provided by ESRI in its ArcMap package.

https://doi.org/10.1371/journal.pntd.0006637.g001

Infection, dissemination and transmission analysis

For each population tested, more than 20 females were analyzed at 6, 9, 14, 21 days post-infection (dpi) and scored infected or non-infected. For each mosquito, abdomen and thorax were analyzed to determine the infection, the head was used for the dissemination and the saliva was tested for the transmission. For all Aedes population tested, transmission efficiencies were obtained by the number of infected saliva divided by the total number of mosquitoes tested for each population at each dpi as previously described [42, 43]. For each saliva collection, females were cold anesthetized before removing their legs and wings. The proboscis was inserted into a filter tip ART (Molecular BioProducts) containing 5 µl of FBS for salivation during 20 minutes. The body and the head were individually ground in 250 µL of DMEM medium supplemented with 2% FBS. Lysis was carried out during 30 sec at 6,000 rpm and the samples were centrifuged at 10,000g during 10 min at 4°C. The supernatants were stored at -80°C before analysis. For the viral detection, ground samples serially diluted were inoculated onto Vero E6 cells in 96-well plates, incubated at 37°C during 7 days and stained with a solution of crystal violet (0.2% in 10% formaldehyde and 20% ethanol). Presence of viral particles was determined by the presence of cytopathic effect (CPE). The saliva samples were stored at -80°C. For detection and titration of ZIKV, saliva samples were inoculated onto Vero E6 cells in 6-wells plates under an agarose overlay and incubated at 37°C during 7 days. Presence of infectious particles was assessed by the detection of plaque and titers were expressed as pfu (plaque-forming unit)/saliva.

Statistical analysis

All rates were statistically compared using Fisher's exact test (R v. 3.3.1) [44], considering p-values > 0.05 non-significant.

Ethics statement

This study follows the New Caledonia Animal Ethics Guidelines.

Results

Different populations of *Aedes aegypti* from the Pacific show similar ZIKV transmission profiles

The ZIKV infection rates of the French Polynesian and the New Caledonian populations of *Ae. aegypti* appeared to be moderate to high (> 53%) (S1 Table and Fig 2A). Significant differences were found between these two mosquito populations at 6 dpi (French Polynesia 53% and New Caledonia 87%, p = 0.009) and 9 dpi (French Polynesia 93% and New Caledonia 73%, p = 0.04). The ZIKV infection rate was significantly lower for the Samoan *Ae. aegypti* during all the course of infection (with a minimal value of 23% at 9 dpi and a maximal value of 50% at 14 dpi).

The dissemination rates were comparable between the three *Ae. aegypti* populations, although the population of Samoa started to be infected only from 14 dpi (S1 Table and Fig 2B). The dissemination rates of *Ae. aegypti* increased during all the course of infection and reached moderate to high levels at 21 dpi (between 53% and 78%).

Infectious viral particles were found in the saliva from 9 dpi for *Ae. aegypti* from New Caledonia. The transmission rates for the three populations tested were low and homogenous (< 30% at 21 dpi) (S1 Table and Fig 2C). The transmission efficiencies of these populations were low, from 3% for the Caledonian population (at 9 dpi) to 6% for the Samoan population and 17% for the French Polynesian one (at 21 dpi), and no significant difference was found (S1



Fig 2. Aedes aegypti from the Pacific region infected with ZIKV. (A) Infection rate, (B) dissemination rate, (C) transmission rate and (D) transmission efficiency at 6, 9, 14 and 21 days post-infection (dpi). Error bars represent 95% confidence intervals. Numbers of mosquitoes tested are indicated above each bar plot. Significant differences are indicated by asterisks (*p < 0.05; **p < 0.01; *** p < 0.001). NT indicates that females were not tested for this analysis point.

https://doi.org/10.1371/journal.pntd.0006637.g002

<u>Table</u> and <u>Fig 2D</u>). However, the extrinsic incubation period seemed shorter for the Caledonian mosquitoes (9 dpi) compared to the two other populations (> 9 dpi).

Low and homogenous transmission of ZIKV for *Aedes polynesiensis* from the Pacific region

The ZIKV infection rates appeared to be high and homogenous from 9 dpi for both the Wallisian and French Polynesian populations of *Ae. polynesiensis* (S2 Table and Fig 3A). At 6 dpi, ZIKV infection rate of the Wallisian vector was higher than that of the Polynesian population (84% vs 23%, p < 0.001).

The ZIKV dissemination rates increased gradually in both vector populations (S2 Table and Fig 3) with no significant difference, except at 21 dpi (p = 0.0135). Although the infection rates were high, dissemination rates were moderate, even at 21 dpi.

Infectious particles in the saliva were found from 14 dpi. Transmission rates were low for the two *Ae. polynesiensis* populations (< 10% at 21 dpi) (S2 Table and Fig 3C). As for *Ae. aegypti* populations from the Pacific region, the ZIKV transmission efficiency values for *Ae. polynesiensis* were low, 2% and 3% (from 14 dpi) for the studied populations (S2 Table and Fig 3D).

Finally, when comparing both results obtained for each population of *Ae. aegypti* and *Ae. polynesiensis*, we observed significant differences between the population of *Ae. aegypti* from Samoa and both *Ae. polynesiensis* populations after 9 dpi for ZIKV infection. Concerning the ZIKV dissemination, few significant differences were observed for the last analysis point (21 dpi) between *Ae. aegypti* from New Caledonia and *Ae. polynesiensis* from Wallis (p = 0.044) and between *Ae. aegypti* and *Ae. polynesiensis* from French Polynesia (p = 0.041). However, no significant difference was found between the two mosquito species for ZIKV transmission.





Fig 3. *Aedes polynesiensis* from the Pacific region infected with ZIKV. (A) Infection rate, (B) dissemination rate, (C) transmission rate and (D) transmission efficiency at 6, 9, 14 and 21 days post-infection (dpi). Error bars represent 95% confidence intervals. Numbers of mosquitoes tested are indicated above each bar plot. Significant differences are indicated by asterisks (*p < 0.05; **p < 0.01; *** p < 0.001).

https://doi.org/10.1371/journal.pntd.0006637.g003

Discussion

Recently, the tropical islands, especially of the Pacific region, have been incriminated as new hubs for arboviruses emergence [45]. Indeed the ZIKV outbreak in 2013 emerged in French Polynesia and subsequently spread through the Pacific region and to Latin America [5, 6, 19]. In this context, the evaluation of *Aedes* vector competence for the arboviruses that circulated in this region appeared to be pivotal. Recent studies estimated the vector competence of the Pacific vectors for different arboviruses at the island scale for dengue virus in New Caledonia and French Polynesia [42], for chikungunya virus in New Caledonia, French Polynesia, and Yap island [27, 46, 47] and for ZIKV in Yap island and French Polynesia [27, 37].

The quick spread of ZIKV in the Pacific region subsequently to the French Polynesian outbreak highlighted the necessity to evaluate the ZIKV vector competence at the regional scale. This study is the first to describe and compare the ZIKV vector competence of *Ae. aegypti* and *Ae. polynesiensis* populations from different Pacific islands for a ZIKV strain which circulated in the region during the 2013–2015 outbreaks.

Our results confirmed that *Ae. aegypti* populations from New Caledonia, Samoa and French Polynesia can transmit ZIKV but with low transmission efficiencies (< 17% until 21 dpi). In addition, even if the infection rates seemed to indicate a better infection of the New Caledonian and the French Polynesian *Ae. aegypti* populations, the transmission efficiency was similar for all populations tested according to our tests. Similar results were obtained in the present study for *Ae. polynesiensis* with infectious viral particles found in the two populations tested, thus demonstrating the potential role of this mosquito species in transmitting ZIKV (< 3% until 21 dpi). For *Ae. aegypti*, the infection rates seemed to indicate heterogeneous profiles, this profile was more homogeneous for *Ae. polynesiensis*. The Samoan population of *Ae. aegypti* and the French Polynesian population of *Ae. polynesiensis* seemed to be less susceptible to the infection

than the populations from the other study sites. A previous study performed in French Polynesia with Ae. aegypti showed higher ZIKV transmission rates (3, 8, 36 and 73% at 6, 9, 14 and 21 dpi) [37]. Such discrepancies may be explained by differences in the experimental protocol: the previous competence work was performed using F16 to F18 generation mosquitoes; moreover, the detection of infectious particles in mosquito saliva was performed on C6/36 instead of Vero cells. The fact that previous experiments were conducted using the French Polynesian instead of New Caledonian ZIKV strain may also be another possible explanation to such dissimilarity, but in both studies the vector competency for ZIKV was confirmed. In contrast to DENV that was previously found as highly disseminating in the New Caledonian and French Polynesian Ae. aegypti, the dissemination rates for ZIKV indicated limited escape from the midgut barrier [42]. Finally, Australian Ae. aegypti has been recently tested for ZIKV transmission [35]. The results showed high prevalence of virus in the saliva. However, the ZIKV strains used although belonging to the Asian lineage was isolated in Cambodia in 2010 (before ZIKV outbreak) thus not harboring the specific amino acids substitution of the Asian American or Pacific sublineage [18, 19]. Differences in the susceptibilities to ZIKV depending on the mosquito species and population, suggest that the outcome of mosquito infection may vary depending on virus-mosquito pairing described as genotype-by-genotype interaction [48]. Moreover, the efficiency of ZIKV in overcoming the midgut infection and escape barrier may be influenced by several factors, notably the microbiome [49, 50] and innate immunity [51, 52] which could be both related to the breeding site environment [53, 54]. The role of the salivary glands as a barrier seemed to be observed for all the populations tested.

Since the explosive ZIKV outbreaks, the vector competence of several mosquito species and populations for this virus has been tested. Previous results highlighted some heterogeneity in the ZIKV transmission, especially due to the ZIKV lineage [55]. In this study, the aim of the experiments was to test the vector competence of Pacific mosquitoes for the ZIKV which circulated during the outbreak (Asian/Pacific lineage). Our results corroborate previous findings that mosquito vectors, whatever the area they originate from (Pacific, America, Europe), are mostly poorly competent for the Asian/Pacific sublineage of ZIKV [2, 36, 37]. We did not have the opportunity here to test Ae. albopictus from the Pacific region. Although, Ae. albopictus could be relevant to Public Health as a potential invasive species in the region, it might not have sustained an outbreak by its own as it is not present in New Caledonia and French Polynesia [22] and as it appears to be a poor ZIKV vector in other region [55]. All together, these observations support that the emergence of ZIKV in the Pacific region and other regions, did not solely rely on the competence of the vectors to transmit this virus. Several other factors, notably the vector density, the mosquito lifespan and non-entomological parameters like the large immunologically naive fractions of the population may have contributed to the unexpected globalization of ZIKV. This work highlights the necessity to characterize the interaction between mosquitoes, arboviruses and humans in the areas recently affected in order to better assess the risk for such arboviruses to emerge and to limit the burden of future outbreaks.

Supporting information

S1 Table. Infection, dissemination, transmission rates and transmission efficiency at 6, 9, 14 and 21 days post-infection (dpi) for *Aedes aegypti* Pacific populations. (DOCX)

S2 Table. Infection, dissemination, transmission rates and transmission efficiency at 6, 9, 14 and 21 days post-infection (dpi) for *Aedes polynesiensis* Pacific populations. (DOCX)

Acknowledgments

We thank the Samoa Ministry of Health for mosquito eggs sampling during entomological investigation in Apia, Western Samoa. We thank Atoloto Malau and the Environmental Services Department (Territory of the Wallis and Futuna Islands) for providing logistical and technical assistance for mosquito sampling in Wallis. We thank Morgane Pol, Sosiasi Kilama (Institut Pasteur de Nouvelle-Calédonie), and Vaea Richard (Institut Louis Malardé) for the mosquito sampling in New Caledonia and French Polynesia respectively. We thank Dinair Couto-Lima (Instituto Oswaldo Cruz—FIOCRUZ) for technical support. We are very grateful to Arnaud Tarantola (Institut Pasteur de Nouvelle-Calédonie) for the mapping.

Author Contributions

Conceptualization: Anna-Bella Failloux, Myrielle Dupont-Rouzeyrol.

Formal analysis: Elodie Calvez.

Funding acquisition: Françoise Mathieu-Daudé, Anna-Bella Failloux, Myrielle Dupont-Rouzeyrol.

Investigation: Elodie Calvez, Laurence Mousson, Marie Vazeille.

Methodology: Elodie Calvez, Anna-Bella Failloux, Myrielle Dupont-Rouzeyrol.

Project administration: Myrielle Dupont-Rouzeyrol.

Resources: Olivia O'Connor, Van-Mai Cao-Lormeau, Françoise Mathieu-Daudé, Nicolas Pocquet.

Supervision: Anna-Bella Failloux, Myrielle Dupont-Rouzeyrol.

Visualization: Elodie Calvez.

Writing - original draft: Elodie Calvez, Myrielle Dupont-Rouzeyrol.

Writing – review & editing: Elodie Calvez, Marie Vazeille, Olivia O'Connor, Van-Mai Cao-Lormeau, Françoise Mathieu-Daudé, Nicolas Pocquet, Anna-Bella Failloux, Myrielle Dupont-Rouzeyrol.

References

- 1. Kuno G, Chang GJ, Tsuchiya KR, Karabatsos N, Cropp CB. Phylogeny of the genus *Flavivirus*. Journal of virology. 1998; 72(1):73–83. PMID: 9420202
- 2. Chouin-Carneiro T, Vega-Rua A, Vazeille M, Yebakima A, Girod R, Goindin D, et al. Differential Susceptibilities of *Aedes aegypti* and *Aedes albopictus* from the Americas to Zika Virus. PLOS Negl Trop Dis. 2016; 10(3):e0004543. https://doi.org/10.1371/journal.pntd.0004543 PMID: 26938868
- 3. Diallo D, Sall AA, Diagne CT, Faye O, Ba Y, Hanley KA, et al. Zika virus emergence in mosquitoes in southeastern Senegal, 2011. PLoS One. 2014; 9(10).
- Duffy MR, Chen T-H, Hancock WT, Powers AM, Kool JL, Lanciotti RS, et al. Zika Virus Outbreak on Yap Island, Federated States of Micronesia. N Engl J Med. 2009; 360(24):2536–43. <u>https://doi.org/10.1056/NEJMoa0805715</u> PMID: 19516034.
- Petersen LR, Jamieson DJ, Powers AM, Honein MA. Zika Virus. N Engl J Med. 2016; 374(16):1552–63. https://doi.org/10.1056/NEJMra1602113 PMID: 27028561.
- Cao-Lormeau VM, Roche C, Teissier A, Robin E, Berry AL, Mallet HP, et al. Zika virus, French polynesia, South pacific, 2013: Emerg Infect Dis. 2014 Jun; 20(6):1085–6. https://doi.org/10.3201/eid2006. 140138; 2014. PMID: 24856001
- Dupont-Rouzeyrol M, O'Connor O, Calvez E, Daures M, John M, Grangeon JP, et al. Co-infection with Zika and dengue viruses in 2 patients, New Caledonia, 2014: Emerg Infect Dis. 2015 Feb; 21(2):381–2. https://doi.org/10.3201/eid2102.141553; 2015. PMID: 25625687

- 8. Kindhauser MK, Allen T, Frank V, Santhana RS, Dye C. Zika: the origin and spread of a mosquito-borne virus. Bull World Health Organ. 2016; 94(9):9.
- Gubio SC, Antonio CB, Silvia IS. Zika Virus Outbreak, Bahia, Brazil. Emerg Infect Dis. 2015; 21 (10):1885. https://doi.org/10.3201/eid2110.150847 PMID: 26401719
- Cao-Lormeau VM, Blake A, Mons S, Lastere S, Roche C, Vanhomwegen J, et al. Guillain-Barre Syndrome outbreak associated with Zika virus infection in French Polynesia: a case-control study. Lancet. 2016; 387(10027):1531–9. https://doi.org/10.1016/S0140-6736(16)00562-6 PMID: 26948433
- Simon O, Acket B, Forfait C, Girault D, Gourinat AC, Millon P, et al. Zika virus outbreak in New Caledonia and Guillain-Barre syndrome: a case-control study. Journal of neurovirology. 2018. Epub 2018/03/ 30. https://doi.org/10.1007/s13365-018-0621-9 PMID: 29594985.
- Igor ADP, Ana Paula PBP, Cristiane WC, Mariana K, Monaise MOS, Lance AW, et al. Time Lags between Exanthematous Illness Attributed to Zika Virus, Guillain-Barré Syndrome, and Microcephaly, Salvador, Brazil. Emerg Infect Dis. 2016; 22(8):1438. https://doi.org/10.3201/eid2208.160496 PMID: 27144515
- Molko N, Simon O, Guyon D, Biron A, Dupont-Rouzeyrol M, Gourinat AC. Zika virus infection and myasthenia gravis: report of 2 cases. Neurology. 2017; 88(11):1097–8. <u>https://doi.org/10.1212/WNL.</u> 00000000003697 PMID: 28188305
- Schuler-Faccini L, Ribeiro EM, Feitosa IM, Horovitz DD, Cavalcanti DP, Pessoa A, et al. Possible Association Between Zika Virus Infection and Microcephaly—Brazil, 2015. MMWR Morb Mortal Wkly Rep. 2016; 65(3):59–62. https://doi.org/10.15585/mmwr.mm6503e2 PMID: 26820244
- Cauchemez S, Besnard M, Bompard P, Dub T, Guillemette-Artur P, Eyrolle-Guignot D, et al. Association between Zika virus and microcephaly in French Polynesia, 2013–15: a retrospective study. Lancet. 2016; 387(10033):2125–32. https://doi.org/10.1016/S0140-6736(16)00651-6 PMID: 26993883
- Faye O, Freire CC, Iamarino A, de Oliveira JV, Diallo M, Zanotto PM, et al. Molecular evolution of Zika virus during its emergence in the 20(th) century. PLOS Negl Trop Dis. 2014; 8(1).
- Lanciotti RS, Lambert AJ, Holodniy M, Saavedra S, Signor Ldel C. Phylogeny of Zika Virus in Western Hemisphere, 2015: Emerg Infect Dis. 2016 May; 22(5):933–5. https://doi.org/10.3201/eid2205.160065; 2016. PMID: 27088323
- 18. Dupont-Rouzeyrol M, Diancourt L, Calvez E, Vandenbogaert M, O'Connor O, Teissier A, et al. Zika virus evolution on the edges of the Pacific ocean. Emerg Microbes Infect. 2017; 6(12):102.
- Pettersson JH, Eldholm V, Seligman SJ, Lundkvist A, Falconar AK, Gaunt MW, et al. How Did Zika Virus Emerge in the Pacific Islands and Latin America? MBio. 2016; 7(5):01239–16.
- Aubry M, Teissier A, Huart M, Merceron S, Vanhomwegen J, Roche C, et al. Zika Virus Seroprevalence, French Polynesia, 2014–2015. Emerg Infect Dis. 2017; 23(4):669. https://doi.org/10.3201/eid2304. 161549 PMID: 28084987
- Gourinat AC, O'Connor O, Calvez E, Goarant C, Dupont-Rouzeyrol M. Detection of Zika virus in urine. Emerg Infect Dis. 2015; 21(1):84–6. https://doi.org/10.3201/eid2101.140894 PMID: 25530324
- 22. Roth A, Mercier A, Lepers C, Hoy D, Duituturaga S, Benyon E, et al. Concurrent outbreaks of dengue, chikungunya and Zika virus infections—an unprecedented epidemic wave of mosquito-borne viruses in the Pacific 2012–2014. Euro Surveill. 2014; 19(41):20929. PMID: 25345518
- Guillaumot L. Arboviruses and their vectors in the Pacific—status report. Pacific Health Dialog. 2005; 12 (2):45–52. PubMed Central PMCID: PMCPMID: 18181493. PMID: 18181493
- Guillaumot L. Entomological survey on Efate—Vanuatu to assess the statut of the mosquito Aedes albopictus. 2013;Doc. n°41/2013-IPNC-URE-EM/LG/DG, 7th of February 2013.
- Guillaumot L, Ofanoa R, Swillen L, Singh N, Bossin HC, Schaffner F. Distribution of Aedes albopictus (Diptera, Culicidae) in southwestern Pacific countries, with a first report from the Kingdom of Tonga. Parasit Vectors. 2012; 5(247):1756–3305.
- 26. Belkin JN. The mosquitoes of the South Pacific: Diptera, Culicidae. Berkeley, University of California Press 1962;Vol. 1.
- 27. Ledermann JP, Guillaumot L, Yug L, Saweyog SC, Tided M, Machieng P, et al. *Aedes hensilli* as a potential vector of Chikungunya and Zika viruses. PLOS Negl Trop Dis. 2014; 8(10).
- Moore PR, Johnson PH, Smith GA, Ritchie SA, Van Den Hurk AF. Infection and dissemination of dengue virus type 2 in *Aedes aegypti, Aedes albopictus*, and *Aedes scutellaris* from the Torres Strait, Australia. J Am Mosq Control Assoc. 2007; 23(4):383–8. https://doi.org/10.2987/5598.1 PMID: 18240514
- Diagne CT, Diallo D, Faye O, Ba Y, Gaye A, Dia I, et al. Potential of selected Senegalese Aedes spp. mosquitoes (*Diptera: Culicidae*) to transmit Zika virus. BMC Infect Dis. 2015; 15(492):015–1231.
- Li MI, Wong PS, Ng LC, Tan CH. Oral susceptibility of Singapore Aedes (Stegomyia) aegypti (Linnaeus) to Zika virus. PLoS Negl Trop Dis. 2012; 6(8):28.

- Wong PS, Li MZ, Chong CS, Ng LC, Tan CH. Aedes (Stegomyia) albopictus (Skuse): a potential vector of Zika virus in Singapore. PLoS Negl Trop Dis. 2013; 7(8).
- Weger-Lucarelli J, Ruckert C, Chotiwan N, Nguyen C, Garcia Luna SM, Fauver JR, et al. Vector Competence of American Mosquitoes for Three Strains of Zika Virus. PLOS Negl Trop Dis. 2016; 10(10).
- **33.** Roundy CM, Azar SR, Rossi SL, Huang JH, Leal G, Yun R, et al. Variation in *Aedes aegypti* Mosquito Competence for Zika Virus Transmission. Emerg Infect Dis. 2017; 23(4):625–32. https://doi.org/10. 3201/eid2304.161484 PMID: 28287375
- Ciota AT, Bialosuknia SM, Zink SD, Brecher M, Ehrbar DJ, Morrissette MN, et al. Effects of Zika Virus Strain and Aedes Mosquito Species on Vector Competence. Emerg Infect Dis. 2017; 23(7):161633.
- **35.** Duchemin J-B, Mee PT, Lynch SE, Vedururu R, Trinidad L, Paradkar P. Zika vector transmission risk in temperate Australia: a vector competence study. Virology journal. 2017; 14(1):108. https://doi.org/10. 1186/s12985-017-0772-y PMID: 28599659
- Jupille H, Seixas G, Mousson L, Sousa CA, Failloux AB. Zika Virus, a New Threat for Europe? PLoS Negl Trop Dis. 2016; 10(8).
- Richard V, Paoaafaite T, Cao-Lormeau V-M. Vector Competence of French Polynesian Aedes aegypti and Aedes polynesiensis for Zika Virus. PLoS Negl Trop Dis. 2016; 10(9):e0005024. <u>https://doi.org/10. 1371/journal.pntd.0005024</u> PMID: <u>27654962</u>
- Calvez E, Guillaumot L, Millet L, Marie J, Bossin H, Rama V, et al. Genetic Diversity and Phylogeny of Aedes aegypti, the Main Arbovirus Vector in the Pacific. PLOS Negl Trop Dis. 2016; 10(1).
- Tabachnick W. Genetics of Insect Vector Competence for Arboviruses. In: Harris K, editor. Advances in Disease Vector Research. Advances in Disease Vector Research. 10: Springer New York; 1994. p. p 93–108.
- Tabachnick WJ. Nature, nurture and evolution of intra-species variation in mosquito arbovirus transmission competence. Int J Environ Res Public Health. 2013; 10(1):249–77. <u>https://doi.org/10.3390/ ijerph10010249 PMID: 23343982</u>
- **41.** Lambrechts L. Quantitative genetics of *Aedes aegypti* vector competence for dengue viruses: towards a new paradigm? Trends Parasitol. 2011; 27(3):111–4. https://doi.org/10.1016/j.pt.2010.12.001 PMID: 21215699
- Calvez E, Guillaumot L, Girault D, Richard V, O'Connor O, Paoaafaite T, et al. Dengue-1 virus and vector competence of *Aedes aegypti* (Diptera: Culicidae) populations from New Caledonia. Parasit Vectors. 2017; 10(1):381. https://doi.org/10.1186/s13071-017-2319-x PMID: 28793920
- 43. Vega-Rua A, Zouache K, Girod R, Failloux AB, Lourenco-de-Oliveira R. High level of vector competence of *Aedes aegypti* and *Aedes albopictus* from ten American countries as a crucial factor in the spread of Chikungunya virus. Journal of virology. 2014; 88(11):6294–306. <u>https://doi.org/10.1128/JVI.</u> 00370-14 PMID: 24672026
- 44. R Foundation for Statistical Computing V, Austria. R Core Team. R: A language and environment for statistical computing. 2016.
- 45. Cao-Lormeau VM. Tropical Islands as New Hubs for Emerging Arboviruses. 2016.
- 46. Dupont-Rouzeyrol M, Caro V, Guillaumot L, Vazeille M, D'Ortenzio E, Thiberge JM, et al. Chikungunya virus and the mosquito vector Aedes aegypti in New Caledonia (South Pacific Region). Vector Borne Zoonotic Dis. 2012; 12(12):1036–41. https://doi.org/10.1089/vbz.2011.0937 PMID: 23167500
- Richard V, Paoaafaite T, Cao-Lormeau V-M. Vector Competence of Aedes aegypti and Aedes polynesiensis Populations from French Polynesia for Chikungunya Virus. PLoS Negl Trop Dis. 2016; 10(5): e0004694. https://doi.org/10.1371/journal.pntd.0004694 PubMed PMID: PMC4856362. PMID: 27144888
- Lambrechts L, Chevillon C, Albright RG, Thaisomboonsuk B, Richardson JH, Jarman RG, et al. Genetic specificity and potential for local adaptation between dengue viruses and mosquito vectors. BMC Evol Biol. 2009; 9(160):1471–2148.
- 49. Apte-Deshpande A, Paingankar M, Gokhale MD, Deobagkar DN. Serratia odorifera a Midgut Inhabitant of Aedes aegypti Mosquito Enhances Its Susceptibility to Dengue-2 Virus. PLoS One. 2012; 7(7): e40401. https://doi.org/10.1371/journal.pone.0040401 PMID: 22848375
- 50. Ramirez JL, Short SM, Bahia AC, Saraiva RG, Dong Y, Kang S, et al. Chromobacterium Csp_P Reduces Malaria and Dengue Infection in Vector Mosquitoes and Has Entomopathogenic and In Vitro Anti-pathogen Activities. PLOS Path. 2014; 10(10):e1004398. https://doi.org/10.1371/journal.ppat. 1004398 PMID: 25340821
- Blair CD, Olson KE. The Role of RNA Interference (RNAi) in Arbovirus-Vector Interactions. Viruses. 2015; 7(2):820–43. https://doi.org/10.3390/v7020820 PubMed PMID: PMC4353918. PMID: 25690800
- Sim S, Jupatanakul N, Dimopoulos G. Mosquito Immunity against Arboviruses. Viruses. 2014; 6 (11):4479–504. https://doi.org/10.3390/v6114479 PubMed PMID: PMC4246235. PMID: 25415198

- 53. Minard G, Mavingui P, Moro CV. Diversity and function of bacterial microbiota in the mosquito holobiont. Parasit Vectors. 2013; 6:146–. <u>https://doi.org/10.1186/1756-3305-6-146</u> PubMed PMID: PMC3667145. PMID: 23688194
- Dickson LB, Jiolle D, Minard G, Moltini-Conclois I, Volant S, Ghozlane A, et al. Carryover effects of larval exposure to different environmental bacteria drive adult trait variation in a mosquito vector. Sci Adv. 2017; 3(8):e1700585. <u>https://doi.org/10.1126/sciadv.1700585</u> PubMed PMID: PMC5559213. PMID: 28835919
- 55. Boyer S, Calvez E, Chouin-Carneiro T, Diallo D, Anna-Bella F. An overview of mosquito vectors of Zika virus. Microbes Infect. 2018; 23(18):006.